DRAFT_2
EO-1 Weekly Status Week of April 14–April 20, 2011
Day of Year 104 - 110
Mission Day 3807 – 3813
Earth Observing One (EO-1) - General
Scheduled 127 science Data Collection Events (DCEs) this week.  
INSTRUMENTS

All instruments operated nominally this week 
· Performed instrument calibration
· Conducted ALI Internal Calibration Type II on April 18 at 00:00z

· Performed Lunar Calibration

· Performed the Nominal Lunar Calibration on April 18 11:20z

Command and Data Handling (C&DH)

EO-1 has continued to experience problems playing back engineering data from the Solid-State Recorders.  Engineering data is being received during real-time contacts and for all science imaging events. 

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.
On Tuesday, April 19, Dan Mandl gave a presentation on the SensorWeb Toolbox software system to an evaluation panel chaired by Scott Green, Code 583.  This panel is evaluating GSFC candidates for the NASA Software of the Year Award.  Dan’s presentation was very well received with no significant negative comments expressed.  If this software is selected as the GSFC candidate, a final presentation will be made to a panel that will select the winner of the NASA Software of the Year Award.
Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week. 
EO-1 MISSION OPERATIONS CENTER

Real-Time

· Continued nominal spacecraft state of health

· Maintaining level 0 data processing and memory handling on data processing computers until our data processing engineer returns

· Returning to work now

· Setting up engineering passes with MGS and SG-1 to test ground stations after maintenance

· MGS certification supports went nominally

· SG-1 tests went nominally, continuing to monitor SG1 passes

Mission Planning

· Created all of the mission planning files for the Lunar Calibration

· Ran them through ASPEN for verification and ASIST SCP comparison

· ASPEN needs a few small tweaks, ASIST SCP came through nominally

· Continued testing of ASPEN to replace the legacy MOPSS mission planning software

· Running ASPEN in conjunction with some normal daily backups from MOPSS

· Running the backup UDAPs through SCP and CMS

Flight Dynamics
No change from previous week.
System Administration
· Continued to maintain EO-1 Data Processing during absence of Tech Engineer.

· Created 2 level 0 archive tapes and 2 dlt tapes for the science community.

· Archived dps user’s mail as a log of at/cron jobs related to processing of scenes.

· Completed updating/implementing password complexity on EO-1 Linux computers to meet the NASA specified requirements.

· Completed applying the April 2011 updates/patches on the 5 EO-1 CNE computers. The patches included Adobe Flash and Microsoft products.

· Completed applying the April 2011 updates/patches on the 3 EO-1 IONet Windows computers. The patches included Adobe Flash and Microsoft products.

· Supported the quarterly All-Highs scan of the EO-1 operational systems performed by the IONet security team. Results of the scan are pending.

· Updated the EO-1 mission firewall in preparation for the TROLL (closed to open IONet proxy server) retirement. Rules were added to allow 4 WGS and 2 SGS computers to send tracking data directly to EO-1 systems. Connection testing will begin shortly.

· Completed update of KACE (patching/reporting tool) to version 5.2 on the 5 EO-1 CNE computers.

· Increased free disk space on the EO-1 backup server by removing backups from 2009. The required 1 year retention policy is still maintained.

· Completed the EO-1 Security Controls equipment list requested by security personnel.

· Sent information regarding the EO-1 Front End Systems to TROLLSAT personnel for the upcoming station tests.

· Replaced the failed monitor on the primary Front End System.
GROUND AND SPACE NETWORK

Station Downtimes

· SG-1 red until further notice (engineering tests scheduled for SG-1, supports working nominally)

· Estimated return to operations 30 days from now

· MGS red for planned maintenance (certification passes ongoing, passes nominal)
Operational Discrepancies
· None
UPCOMING EVENTS

· Inclination maneuvers on hold until further notice

· Next lunar calibration to be mid-May
Imagery Status

Scenes and Engineering Cals planned for week of April 14, 2011-April 20, 2011      127 
Total scenes and engineering calibrations planned for entire mission – approximately 59,646
Total Scenes:  ALI scenes in the Level 0 archive              54,958                (as of April 20, 2011)

                         Hyperion scenes in the Level 0 archive    54,700               
PUBLICATIONS AND PRESENTATIONS STATUS 
430 publications 

295 presentations

53 articles and press releases[image: image1.png]
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